Author’s response to reviews

Title: Limb symmetry during double-leg squats and single-leg squats on land and in water in adults with long-standing unilateral anterior knee pain; a cross sectional study

Authors:

Anna Severin (aseverin@usc.edu.au)
Brendan Burkett (bburkett@usc.edu.au)
Mark McKeen (mmckean@usc.edu.au)
Aaron Wiegand (awiegand@usc.edu.au)
Mark Sayers (msayers@usc.edu.au)

Version: 2 Date: 16 Nov 2017

Author’s response to reviews:

The authors thank the Reviewers for their input and feedback. Please find attached document ('Response to reviewers 2nd') among the uploaded files to see our responses to the specific comments.